Author's response to reviews

Title: Treatment of hyperfunctioning thyroid nodules by percutaneous ethanol injection: 2-year follow up

Authors:

Dr Bagher Larijani (emrc@sina.tums.ac.ir)
Dr Mohammad Pajouhi (emrc@sina.tums.ac.ir)
Dr Hossein Ghanaati (emrc@sina.tums.ac.ir)
Dr Mohammad Hassan Bastanhagh (emrc@sina.tums.ac.ir)
Dr Fereshteh Abbasvandi (emrc@sina.tums.ac.ir)
Dr Karim Firooznia (emrc@sina.tums.ac.ir)
Dr Mahmood Shirzad (emrc@sina.tums.ac.ir)
Dr Mohammad Reza Amini (emrc@sina.tums.ac.ir)
Dr Maryam Saraie (emrc@sina.tums.ac.ir)
Dr Nasrin Abbasvandi (emrc@sina.tums.ac.ir)
Dr Reza Baradar Jalili (emrc@sina.tums.ac.ir)

Version: 3 Date: 5 Dec 2002

PDF covering letter
Dear Ms Sawyerr,

We have applied the formatting changes recommended in your e-mail.

We have also made the discretionary revisions suggested by Professor Monzani, as follows:

1. We have deleted the reference to nodular fibrosis in the abstract
2. We have amended the paragraph on statistical analysis in the methods section to clarify the point that correlation testing was performed to evaluate the relationship between nodule volume and reduction in nodule size. Our response to Professor Monzani’s initial report did not mean that we did not use correlation testing but rather that we do not think repeating it or applying it to other variables would add much to the findings of the study.
3. We have added details of the patients with persistent TSH suppression.

Many thanks for all your help and patience,

Best regards,

Bagher Larijani,
Director and Chief Scientific Officer
Endocrinology & Metabolism Research Centre
Tehran University of Medical Sciences